Author's response to reviews

Title: The early response of renal cell carcinoma to tyrosine kinase inhibitors evaluated by FDG PET/CT was not influenced by metastatic organ

Authors:

Manabu Kakizoe (makakizoe@yahoo.co.jp)
Masahiro Yao (masayao@med.yokohama-cu.ac.jp)
Ukihide Tateishi (utateish@yokohama-cu.ac.jp)
Ryogo Minamimoto (ryogominamimoto@yahoo.co.jp)
Daiki Ueno (daikochi810903@yahoo.co.jp)
kazuhiro Namura (namu_ck@yahoo.co.jp)
Kazuhide Makiyama (makiya@yokohama-cu.ac.jp)
Narihiko Hayashi (twnary@yahoo.co.jp)
Futoshi Sano (fsano@yokohama-cu.ac.jp)
Takeshi Kishida (kishidat@kcch.jp)
Kazuki Kobayashi (kazuki1010@ybb.ne.jp)
Sumio Noguchi (sn0818@kf6.so-net.ne.jp)
Ichiro Ikeda (iichiban@yahoo.co.jp)
Yoshiharu Ohgo (y-oogo@yokohamasakae.jp)
Masaharu Taguri (taguri@yokohama-cu.ac.jp)
Satoshi Morita (smorita@kuhp.kyoto-u.ac.jp)
Tomio Inoue (tomioi@yokohama-cu.ac.jp)
Yoshinobu Kubota (kubotayo@med.yokohama-cu.ac.jp)
Noboru Nakaigawa (nakaigan@med.yokohama-cu.ac.jp)

Version: 4 Date: 21 April 2014

Author's response to reviews:

Dear Editor

I am so glad that the reviewers and senior editor evaluated our report titled “The early response of renal cell carcinoma to tyrosine kinase inhibitors evaluated by FDG PET/CT was not influenced by metastatic organ”. We believe that our information is very important for many clinical urologists.

We answered to the editor comment as below.

Noboru Nakaigawa MD & PhD
Associate Professor
Department of Urology
Yokohama City University School of Medicine

EDITORS COMMENTS:
This manuscript is worthy of publication. However, please make the edit as
requested by both reviewers to include standard deviation and range of uptake times for each PET/CT examination. The authors state that their facility is "punctual", but along with the reviewers I find it hard to believe that each and every scan occurred at exactly 60 minutes post injection.

Answer

We added the description about time from injection to imaging (mean, SD, and range) in Results (p.7, l.12-13).